Author’s response to reviews

Title: An Investigation of the Effects of Infertility on Women's Quality of Life: A Case-Control Study

Authors:
Katayoun Bakhtiar (k_bakhteyar@yahoo.com)
Ramin Beiranvand (beiranvand83@yahoo.com)
Arash Ardalan (arashard@gmail.com)
Farahnaz Changaee (farahnazchangavi@yahoo.com)
Mohammad Almasian (almasian26@gmail.com)
Afsaneh Badrizadeh (badrizadeha@yahoo.com)
fatemeh bastami (fatemeh2011bastami@gmail.com)
Farzad Ebrahimzadeh (farzadebrahimzadeh@gmail.com)

Version: 4 Date: 25 Jul 2019

Author’s response to reviews:

1- authors’ names were represented using their initials.
2- list of all the abbreviations used in the manuscript were provided.
3- my revised manuscript was upload as a single, final